Author’s response to reviews

Title: Superior Mesenteric Artery Syndrome coexists with Nutcracker Syndrome in a female: a case report

Authors:

Ying Shi (pepper-ann@163.com)
Guangya Shi (shiguangya@sina.cn)
Zhenyu Li (274081847@qq.com)
Yanfang Chen (260631930@qq.com)
Shaohui Tang (tangshaohui205@163.co)
Wei Huang (thuangw@163.com)

Version: 1 Date: 24 Aug 2018

Author’s response to reviews:

1. Thank you for reminding. The overlap parts of Background and Case presentation section in the manuscript have been revised.

2. We have supplied the authors’ contributions according to individual contributions.

3. The single and clean manuscript has been revised and uploaded.